Author's response to reviews

Title: Being a quantitative interviewer: qualitatively exploring interviewers' experiences in a longitudinal cohort study

Authors:

Sarah Derrett (sarah.derrett@otago.ac.nz)
Sarah Colhoun (sarah.colhoun@otago.ac.nz)

Version: 3 Date: 8 December 2011

Author's response to reviews: see over
8 December 2011

Dear Editor,

**Title:** Being a quantitative interviewer: qualitatively exploring interviewers’ experiences in a longitudinal cohort study  
**Journal:** BMC Medical Research Methodology MS ID: 4531651846107974

Thank you very much for accepting our paper for publication in your journal. Below are our responses to the requested formatting changes. The uploaded version has been prepared for publication.

Formatting changes to be made:

*-----------------------------*
**Major revisions (we require the author to make these changes)**
References: Please ensure that references are consistently placed before the full stop. For example "text [1]."
Done.
Authors’ contributions: Please indicate in this section that both authors read and approved the final manuscript.
Done.

*-------------*
**Minor revisions** (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)
Abstract: Please rename "Results and Discussion" heading to "Results" in both the manuscript and submissions system
Done.
References: please provide the full citation details for reference 10. You can view the BMC reference guide at the following link
http://www.biomedcentral.com/bmcgenet/authors/instructions/researcharticle#formatting-references
Done.
Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
Done.

*-------------*
**A small change to the proofs:** In the last sentence of the paper’s Conclusion and Abstract the phrase – “…establishing and ‘ethical’ relationship” has become “establishing an ethical relationship” (i.e. we have removed the parentheses around the word ethical).

Yours sincerely
Sarah Derrett